Author's response to reviews

Title: Correlation between Progetto Cuore Risk Score and Early Cardiovascular Damage in Never Treated Subjects

Authors:

Aldo Pende (apende@unige.it)
Caterina Grondona (categrandoma@virgilio.it)
Stefano Bertolini (stefbert@unige.it)

Version: 3 Date: 12 September 2008

Author's response to reviews: see over
To Eugenio Picano
Editor-in-Chief
Cardiovascular Ultrasound

Please find herewith enclosed a final version of our manuscript entitled “Correlation between Progetto Cuore Risk Score and Early Cardiovascular Damage in Never Treated Subjects” with the formatting changes you required.

Best regards, yours sincerely

Aldo Pende, M.D.
Department of Internal Medicine
University of Genoa School of Medicine
Viale Benedetto XV, 6
16132 Genoa, Italy
Tel. & Fax 00390103537524
e-mail: apende@unige.it